Author’s response to reviews

Title: Perceptions of Brazilian Nursing Faculty members regarding Literacy of Human Rights related to Health in Nursing Undergraduate Programs

Authors:

Carla Aparecida Arena Ventura (caaventura@eerp.usp.br)
Isa de Godoy (sig@eerp.usp.br)
Simone de Godoy (sig@eerp.usp.br)
Laís Fumincelli (laiscelli13@gmail.com)
Mirella Castelhano Souza (micasellhano@yahoo.com.br)
Valtuir Souza-Junior (valtuirduarte@usp.br)

Version: 4 Date: 12 Aug 2019

Author’s response to reviews:

Dear Editor,

At this moment, we are not able to share the raw data, based on the National Council of Research Ethics legislation and the University of Sao Paulo rules.

Regarding the questionnaire used, it was previously published and we cite in this manuscript as reference 17: Ventura CAA, Mendes IAC, Wilson LL. et al. Godoy S, Tamí-Maury I, Zárate-Grajales R et al. Global health competencies according to nursing faculty from Brazilian higher education institutions. Rev Latino-Am Enfermagem. 2014;22:179-86; doi: 10.1590/0104-1169.2996.2400.